Author's response to reviews

Title: Meteorologically estimated exposure but not distance predicts asthma symptoms in schoolchildren in the environs of a petrochemical refinery: a cross-sectional study

Authors:

Neil White (rodney.ehrlich@uct.ac.za)
Jim teWaterNaude (jimtewn@mweb.co.za)
Anita van der Walt (anitavdw@mweb.co.za)
Grant Ravenscroft (grant@ecoserv.com)
Wesley Roberts (wroberts@csir.co.za)
Rodney Ehrlich (rodney.ehrlich@uct.ac.za)

Version: 5 Date: 23 August 2009

Author's response to reviews:

Abstract, line 4: "calculated" substituted for the second "derived".
Under Methods: Outcome and covariate exposure measurement, line 3: words "presence of" dropped.
Table 1, Title: "by" substituted for "according to"
Abbreviations, Acknowledgements, References, Tables: a number of spacing, alignment, punctuation and font inconsistencies have been corrected.